



# **Clinical Trials in Pediatric Autosomal Dominant Polycystic Kidney Disease**

#### Melissa A. Cadnapaphornchai\*

University of Colorado Anschutz Medical Campus, Aurora, CO, USA

Autosomal dominant polycystic kidney disease (ADPKD) is the most common hereditary kidney disease and is associated with concerning long-term implications for kidney function and cardiovascular health. Early intervention is needed in order to mitigate these long-term complications. Herein, we review important findings from recent clinical trials in ADPKD and their relevance to affected children and young adults and consider future directions for intervention. Recent clinical trials support aggressive control of blood pressure with blockade of the renin-angiotensin-aldosterone system as well as potential benefit of pravastatin therapy in children and young adults with ADPKD. There are several other candidate therapies, some of which have shown benefit in adult ADPKD, which require further investigation in affected children.

### OPEN ACCESS

#### Edited by:

Erum Aftab Hartung, Children's Hospital of Philadelphia, USA

#### Reviewed by:

Andrew Mallett, Royal Brisbane and Women's Hospital, Australia Hans Van Rostenberghe, Universiti Sains Malaysia, Malaysia

#### \*Correspondence:

Melissa A. Cadnapaphornchai melissa.cadnapaphornchai@ ucdenver.edu

#### Specialty section:

This article was submitted to Pediatric Nephrology, a section of the journal Frontiers in Pediatrics

Received: 20 November 2016 Accepted: 28 February 2017 Published: 23 March 2017

#### Citation:

Cadnapaphornchai MA (2017) Clinical Trials in Pediatric Autosomal Dominant Polycystic Kidney Disease. Front. Pediatr. 5:53. doi: 10.3389/fped.2017.00053 Keywords: autosomal dominant polycystic kidney disease, child, pediatric, clinical trials, therapy

Autosomal dominant polycystic kidney disease (ADPKD) is the most common hereditary kidney disease, affecting one in 500 individuals and accounting for nearly 5% of end-stage kidney disease in the United States (1). Although historically considered a disease of adults, it has become clear in recent decades that ADPKD can have significant clinical manifestations as early as *in utero*. Early intervention to mitigate progression of structural kidney disease is anticipated to have the most long-term impact on outcome. Thus, recent clinical trials have focused on early stage ADPKD and specifically on the potential benefits of intervention during childhood. The purpose of this article is to review important findings from recent clinical trials in children and young adults with ADPKD and to discuss future directions for therapeutic intervention.

## LESSONS LEARNED FROM NATURAL HISTORY STUDIES IN CHILDREN

Observational studies in the 1980s and 1990s in children at risk for ADPKD (i.e., one parent known to have ADPKD) began to delineate the extent of disease in children and to identify specific risk factors for subsequent intervention (2, 3). Through such studies, we learned that children with ADPKD had larger kidneys which grew faster than those of unaffected children. Moreover, the larger the kidneys were at baseline imaging, the faster they grew over time. Ultrasonography was shown to be a reliable non-invasive tool for the diagnosis of ADPKD in children. Specifically with the degree of resolution available in the 1990s, this method could delineate kidney cysts in nearly 80% of children aged 5–18 years with known *PKD1* mutations (2). The presence of bilateral cysts in at-risk children is sufficient for diagnosis (4). Many of the same clinical manifestations, which had typified adult ADPKD, including pain, gross and/or microhematuria, proteinuria, and impaired urinary concentrating ability, were also present in affected children, with frequency of symptoms correlating to extent of structural kidney disease. Early cardiovascular disease was also of concern in the pediatric population, with a high prevalence of hypertension and elevated left ventricular

1

mass index (LVMI) as compared to unaffected children (5, 6). Distinct from adults, however, the vast majority of children with ADPKD maintained normal kidney function despite progressive structural kidney disease, thus emphasizing the great importance of total kidney volume (TKV) as a measureable outcome of early disease progression and the primary focus of therapeutic intervention in the pediatric ADPKD population. Although there is good correlation between ultrasound and magnetic resonance imaging (MRI) assessment of TKV in ADPKD children (7), MRI is the preferred method to follow TKV in the research setting and clearly is a more accurate and reliable measure as structural disease advances (8).

## **BLOOD PRESSURE CONTROL IN ADPKD**

Blood pressure control is a critical aspect of chronic kidney disease (CKD) management in both children and adults, with numerous studies showing delayed deterioration in kidney function with more aggressive control (9–11). The prevalence of hypertension in pediatric ADPKD is estimated to be 20% (12), with an even higher prevalence in very early onset ADPKD (13, 14). Previous studies in adults with ADPKD implicated a major role for the renin-angiotensin-aldosterone system (RAAS) in the pathogenesis of hypertension and left ventricular hypertrophy [reviewed in Ref. (15, 16)]. Although postulated, there has however been no direct evidence that RAAS induces kidney cyst growth in this condition.

With these considerations, an interventional trial was designed to assess the effect of aggressive control of blood pressure with angiotensin-converting enzyme inhibition (ACEI) on kidney and cardiovascular disease progression in children and young adults ages 4-21 years with ADPKD (17). Study participants with hypertension (blood pressure above the 95th percentile for height, sex, and age) were randomized to enalapril treatment with goal blood pressure at the 90th percentile (HBP90) or at the 50th percentile (HBP50), while participants with high normal blood pressure (75th-95th percentile for height, sex, and age) were randomized to either enalapril with goal blood pressure at the 50th percentile (NBP50) or to observation without treatment (NBP90). All participants were followed annually for 5 years with routine laboratory testing, 24-h urine creatinine clearance as an estimate of glomerular filtration rate (eGFR), abdominal ultrasound and MRI assessment of TKV, and echocardiographic assessment of LVMI.

Not surprisingly, participants in the HBP50 group required more antihypertensive medications (mean  $\pm$  SEM: 2.8  $\pm$  0.3 vs. 1.6  $\pm$  0.4; P < 0.05) than HBP90 participants. Regardless of the degree of hypertension control, however, participants in both of these study groups demonstrated a significant increase in TKV and decrease in eGFR over the study period. No significant differences were observed in LVMI or urine albumin excretion between study groups or over time. These results emphasized the high risk for structural and functional kidney disease progression associated with hypertension in children with ADPKD.

The clinical trial further demonstrated that baseline LVMI in high normal blood pressure participants was similar to that of hypertensive participants and was elevated compared to those with blood pressure below the 75th percentile, suggesting that early cardiovascular involvement is present even with blood pressure in the upper quartile of the normal range. Both NBP50 and NBP90 groups demonstrated a rise in TKV over time despite BP control with ACEI in the NBP50 group. However, participants in the NBP90 group demonstrated significant decrease in eGFR and increase in LVMI over the study period, while these parameters were stable in the NBP50 group. No significant difference was observed in urine albumin excretion between these study groups at baseline or over time. These findings suggested that children and young adults with ADPKD and high normal blood pressure may represent a particular subgroup for whom rigorous control of blood pressure with ACEI could confer renal functional and cardiovascular protection.

Limitations of this single-center pediatric research study include the relatively small number of participants as compared to adult trials despite nationwide recruitment, the restricted number of participants consistently achieving goal BP in the hypertensive group (40% in HBP50 vs. 70% in HBP90 while 100% of NBP participants met BP goal), and the generous dropout rate (27% overall). The latter speaks to a common struggle associated with longitudinal clinical trials in the pediatric population (18, 19).

Subsequently, the multicenter HALT-PKD study was conducted (20). This double-blind, placebo-controlled trial included 558 hypertensive participants 15-49 years of age with early stage ADPKD (eGFR greater than 60 mL/min/1.73 m<sup>2</sup>) who were randomized to either a standard blood pressure target (120/70-130/80 mm Hg) or a low blood pressure target (95/60-110/75 mm Hg) and to either the ACEI lisinopril plus an angiotensin-receptor blocker (telmisartan) or to lisinopril plus placebo. Serial follow-up was performed for 5-8 years with a primary outcome of annual percentage change in TKV by MRI. The trial included a very limited number of participants in the pediatric age group, with mean age of study participants overall of  $36 \pm 8$  years. Combined lisinopril/telmisartan did not significantly alter the rate of increase in TKV as compared to lisinopril/placebo. However, as compared with standard blood pressure control, rigorous blood pressure control was associated with a small (1.1% per year) but significant reduction in TKV and no change in eGFR. Not surprisingly, there was a 16% increase in dizziness and light headedness in the low blood pressure target group.

It is important to note that in the HALT-PKD study, there was no control arm without RAAS blockade. Thus the ability to assess the specific effect of RAAS blockade on TKV was limited. In addition, there were more participants in the low BP group with *PKD2* mutations (19.8 vs. 13.1%); *PKD2* is associated with less progressive kidney disease as compared to *PKD1* (21). Finally, the percentage of subjects consistently meeting BP goal was limited. Specifically, the systolic and diastolic blood pressures, as measured at home, were on target across all study visits in 40–66 and 58–75% of participants in the low-blood pressure group, respectively, and in 32–48 and 33–52% of those in the standard-blood pressure group, respectively. These factors could each have affected the outcome of the clinical trial.

The KDIGO Controversy Conference on ADPKD has suggested a target blood pressure of less than 140/90 in adults with ADPKD with a stricter goal below 130/80 if macroalbuminuria is present (22) and further recommended management of hypertension in ADPKD children per routine guidelines for the general pediatric population (goal below 90th percentile for age, sex, and height) (23) with RAAS blockade as the preferred first-line treatment for hypertension in ADPKD (22). Despite these recommendations, the above-described clinical trials intimate that more aggressive control of BP might be of value for both kidney and cardiovascular reasons in select ADPKD patients. Further investigations are needed to better characterize the long-term relevance of these observations.

Hyperkalemia and reduced GFR are unusual in pediatric ADPKD as kidney function is usually normal. However, the risk of fetal birth defects with RAAS blockade during pregnancy remains an important topic for discussion when providing such treatment to females of appropriate pubertal development.

## STATIN THERAPY IN PEDIATRIC ADPKD

HMG-CoA reductase inhibitors (statins) have been shown to enhance renal blood flow and GFR and to attenuate inflammation through vascular and glomerular nitric oxide production [reviewed in Ref. (24, 25)]. In an animal model of ADPKD, lovastatin reduced the severity of structural and functional kidney disease (26), and in a small study of ADPKD adults, short-term (4-week) treatment with simvastatin was associated with improved renal blood flow and GFR (27). Although the underlying mechanisms are not well understood, it has been proposed that these renoprotective effects are mediated by statin-related inhibition of G proteins, leading to decreased cell proliferation.

In this context, a 3-year randomized double-blind placebocontrolled phase III clinical trial of pravastatin treatment on kidney and cardiovascular disease progression in children and young adults ages 8-22 years with ADPKD and normal kidney function was conducted (28). The primary outcome variable was a combined endpoint of  $\geq$ 20% increase in TKV corrected for height (HtTKV), LVMI, or urine albumin excretion over the study period. All participants received ACEI treatment. Fewer participants receiving pravastatin achieved the primary endpoint compared with placebo (69 vs. 88%; P = 0.03) (29). This was due primarily to a lower proportion reaching the increase in HtTKV (46 vs. 68%; P = 0.03), with similar findings observed between study groups for LVMI (25 vs. 38%; P = 0.18) and urine albumin excretion (47 vs. 39%; P = 0.50). The percent change in HtTKV adjusted for age, sex, and hypertension status over the 3-year period was significantly decreased with pravastatin ( $23 \pm 3\%$  vs.  $31 \pm 3\%$ ; P = 0.02). In mixed-model longitudinal data analysis, the increase in percent growth over 3 years was significantly different between the statin and placebo groups (P = 0.01). The medication was well tolerated with no adverse effects on serum liver or muscle enzymes; no participant discontinued pravastatin due to side effects.

The mechanisms of statin effect on kidney cyst growth remain to be fully elucidated. Mass spectrometry-based analysis of biomarkers of endothelial dysfunction, inflammation, and oxidative stress was pursued in this cohort (30). Pravastatin therapy was associated with reduced plasma concentrations of cyclooxygenase- and lipoxygenase-derived plasma lipid mediators (e.g., 9-hydroxyoctadecadienoic acid, 13-hydroxyoctadecadienoic acid, and 15-hydroxyeicosatetraenoic acid) over the course of the research study as compared to placebo. These metabolites have been previously shown to enhance the profibrotic effects of angiotensin II (31) and to activate the peroxisome proliferator-activated receptor  $\gamma$ , which has been implicated in cyst growth (32). Thus reduction *via* statin therapy would be anticipated to be of benefit in ADPKD.

These findings support the use of pravastatin to ameliorate progressive structural kidney disease in children and young adults with ADPKD. Whether statin therapy may be beneficial in earlier childhood (i.e., prior to 8 years of age) or in more advanced ADPKD as in older adults is not currently known and requires study. With a readily available, well-tolerated medication which can mitigate progression of kidney disease in children with ADPKD, however, issues now arise regarding whether/when to screen at-risk children. A thoughtful discussion with families regarding the potential benefits and risks of diagnosis (including potential impact on future life/disability insurance) and implications of treatment on long-term prognosis is necessary. Early intervention remains critical in order to have the most long-term impact on kidney and cardiovascular outcomes (33-35). As with ACEI, potential risks to the fetus with statin exposure during pregnancy should be reviewed as appropriate prior to treatment.

## CLINICAL TRIALS IN ADULTS WITH ADPKD AND IMPLICATIONS FOR PEDIATRIC ADPKD

Results from other recent clinical trials in adults with ADPKD must also be considered, recognizing as always that findings from adult studies are not necessarily generalizable to the pediatric population. In 1989, Grantham et al. reported that in human kidney ADPKD cells (but not normal kidney cells), vasopressin-induced cyclic 3',5'-adenosine monophosphate (cAMP) stimulated chloride-driven fluid secretion, and B-Raf/ Mek/extracellular signal-regulated pathway cell proliferation and dedifferentiation, implicating cAMP as an important regulator of growth in ADPKD kidney cysts (36, 37). Some years later, Gattone et al. demonstrated an impressive decrease in cyst fluid accumulation via blockade of the vasopressin V2 receptor in rat and mouse models of polycystic kidney disease (38, 39). Subsequently, the multicenter international TEMPO 3:4 study included nearly 1,500 adults (aged 18-50 years) with ADPKD, eGFR above 60 mL/min/1.73m<sup>2</sup>, and relatively large kidneys (TKV > 750 mL) who were randomized to vasopressin V<sub>2</sub> receptor blockade with tolvaptan or to placebo for 3 years (40, 41). All participants were advised to avoid thirst by drinking a glass of water after each urination. Participants in the tolvaptan arm demonstrated slower annual increase in TKV than controls (2.8 vs. 5.5%; P < 0.001) and slower annual decline in kidney function (-2.6 vs. -3.8 mg/mL; P < 0.001). Suppression of plasma arginine vasopressin (AVP) concentration was observed in both groups,

likely due to voluntary water intake by placebo participants. Thus the observed magnitude of effect between treatment and placebo groups may have been diminished. It is important to note that the TEMPO clinical trial included participants with relatively advanced disease (TKV > 750 mL) and thus the potential benefit to the general ADPKD population may be overestimated based on trial results. It is further unknown if the effect on TKV and renal functional decline would be sustained with ongoing treatment beyond a 3-year period.

Based on these clinical trial results, tolvaptan was subsequently approved for rapidly progressive ADPKD in adults in Europe, Japan, Korea, and Canada but not in the United States. There is nevertheless some variability regarding the definition of "rapid progression" and thus which patients merit therapy. Although it has been proposed that tolvaptan could decrease annualized TKV growth by 1.99% per year (P < 0.001) and eGFR decline by 0.40 (P = 0.23) in CKD stage 1 (42), for example, European guidelines have recommended against starting tolvaptan in patients aged 30-40 years with CKD stage 1 (eGFR > 90 mL/min/1.73 m<sup>2</sup>) due to slow progression of disease (43). The U.S. Food & Drug Administration denied the new drug application for tolvaptan for ADPKD in 2013, citing concern for side effects (including significant liver enzyme elevation in 4.4%) and high study dropout rate (23 vs. 14%) in the TEMPO trial. Of note, two participants met definition for Hy's Law case although all cases resolved with drug withdrawal. Despite the positive results of the TEMPO trial, concern also remains regarding the cost-effectiveness of tolvaptan for ADPKD, with a recent Markov-based analysis suggesting that while tolvaptan therapy could prolong median age at end-stage kidney disease by 6.5 years, this came at a cost of \$744,100 per quality-adjusted life-year gained (44).

With the positive results of the TEMPO trial, there is great interest to apply tolvaptan to pediatric ADPKD. Several considerations are needed however in planning such a pediatric trial. The severity of structural kidney disease as studied in the TEMPO trial (TKV > 750 mL) is not as common among the general pediatric ADPKD population. Thus, the effect of therapy may be less impressive than in the TEMPO trial, and there are significant potential side effects as detailed above. Furthermore, the vast majority of children with ADPKD maintain normal renal function (CKD stage 1), and a validated method to identify slow vs. rapid progressors does not exist in the pediatric population. Therefore, patient selection criteria must be carefully delineated. Finally, the lower age limit at which children can be expected to appropriately compensate for vaptan-induced polyuria has not been well established. These and other concerns have led to delay in initiation of a pediatric ADPKD clinical trial with tolvaptan although future carefully designed studies should be forthcoming.

It has been proposed that modulation of kidney cyst growth similar to V<sub>2</sub> receptor blockade might be achieved easily and safely with suppression of AVP *via* chronic high fluid intake (45). Such an intervention is attractive for use in the pediatric age group due to its low risk of potential side effects. The feasibility of inducing low urine osmolality has been demonstrated in a small pilot study in ADPKD adults (46). Thirty-four participants ages 18–60 years with ADPKD and eGFR  $\geq$  60 mL/min/1.73 m<sup>2</sup> were randomly assigned to a low-osmolar diet vs. no intervention for a 2-week period. The intervention group demonstrated a significant decrease in urine osmolality and mean plasma copeptin levels, a marker of AVP effect, over the study period. However, a small non-randomized clinical trial of high fluid intake for 1 year in adults with ADPKD was paradoxically associated with increased TKV (47). It remains to be determined whether constant high fluid intake on a prolonged basis is truly practicable in either adults or children, the latter whom may have some difficulty consistently following a low-osmolar diet with protein needs for growth, and thereafter what effect these interventions might have on kidney cyst growth long term.

Somatostatin analogs such as octreotide and lanreotide have been utilized in adults with ADPKD to impact both polycystic liver and polycystic kidney disease. These medications inhibit cAMP production through stimulation of G $\alpha$ i. The ALADIN study demonstrated diminished growth in TKV after 1 year of treatment with octreotide but the effect was not sustained after 3 years (48, 49). An observational study of 6-month treatment with lanreotide in adults with polycystic liver/kidney disease demonstrated a slight but statistically significant decrease in median TKV (1,023–1,012 mL; P = 0.006) (50). Further study is needed to determine whether there is a sustained clinically significant effect of these medications on TKV in ADPKD. Such medications may have a more important role in management of PKD-associated polycystic liver disease, which would be an exceedingly rare occurrence in childhood ADPKD.

Inhibitors of the mammalian target of rapamycin (mTOR) have been previously studied in adult ADPKD. The mTOR pathway plays an important role in cell growth and proliferation, including interaction of the cytoplasmic tail of polycystin 1 with tuberin. Inappropriate activation has been observed in some cyst-lining epithelial cells in human ADPKD patients and in mouse models of ADPKD (51, 52). Animal studies suggested that rapamycin, an inhibitor of mTOR, reduced cystogenesis, an effect which is dependent on appropriate blood levels and the tubular origin of cyst development (52, 53). Such studies led to subsequent clinical trials of mTOR inhibition in adult ADPKD with disappointing results. Sirolimus treatment had no effect on TKV or kidney function over an 18-month period in adults with ADPKD and normal kidney function (54). Gastrointestinal side effects of sirolimus were common, with 82% of treated participants complaining of mucositis. Everolimus treatment for 2 years in adults with ADPKD resulted in a non-significant slower rate of increase in TKV and faster decline in eGFR (55). Cytopenias and mucositis occurred significantly more frequently in the treatment group. These medications have not been studied in pediatric ADPKD, but their use is likely to be limited by the high rate of side effects and concern for possible deleterious effects of long-term immunosuppression beginning early in life.

An alternative approach for suppressing mTOR activity is to utilize agents that activate AMP-activated kinase (AMPK). This kinase has been shown to antagonize mTOR activation (56). In this regard, metformin, a pharmacological activator of AMPK, has been demonstrated to arrest cyst growth in mouse models of PKD (57). Metformin has the advantage of being extensively utilized in clinical and research settings in a variety of medical conditions including diabetes and polycystic ovary syndrome in both adults and children. Thus, ample safety data are available for review. The medication is also relatively inexpensive and may have anti-inflammatory and anti-fibrotic effects [reviewed in Ref. (58)], which could be of potential benefit for vascular health in ADPKD. Clinical trials are currently ongoing to assess the effect of metformin on kidney cyst growth and function in adults with ADPKD (NCT02903511, NCT02656017). With increasing experience with metformin in children with various medical conditions, metformin represents a candidate intervention for children with ADPKD which requires further reflection.

Curcumin, from the plant curcuma longa, is a naturally occurring polyphenol found in the spice turmeric, which has the unique ability to activate transcription of key antioxidants, suppress inflammation, and reduce cell proliferation, leading to recent application in conditions of abnormal cell growth including cancer. In various animal models of acute and chronic kidney injury, curcumin treatment has been shown to mitigate renal injury (59-61), possibly by reduced activation of the proinflammatory transcription factor NFkB and its downstream target tumor necrosis factor- $\alpha$  (TNF- $\alpha$ ), concomitant with an increase in the anti-inflammatory transcription factor peroxisome proliferator-activated receptor-y (PPARy) (62, 63). Curcumin has been shown to slow cyst growth in vitro in a dose-response manner, using both the Madin-Darby canine kidney cell cyst model and an embryonic kidney cyst model (64). These effects appear at least in part to be mediated by alteration in intracellular signaling proteins like Ras, B-raf, p-MEK, p-ERK, c-fos, and Egr-1. In the Pkd-1 deletion mouse model, curcumin improves renal histology and reduces proliferative index, cystic index, and kidney weight normalized to total body weight (65). Curcumin has also been shown to reduce vascular dysfunction in rodent models of hypertension, diabetes, and aging (66-69). These improvements are associated with increased vascular nitric oxide bioavailability and arterial eNOS expression, reduced vascular oxidative stress (decreased reactive oxygen species, reduced oxidative damage, and increased antioxidant enzymes), and increased activity of hemoxygenase-1, which can promote anti-inflammatory pathways (66-69). Vascular dysfunction is known to occur in adults with ADPKD (70), and cardiovascular disease is the leading cause

## REFERENCES

- Saran R, Li Y, Robinson B, Ayanian J, Balkrishnan R, Bragg-Gresham J, et al. US renal data system 2014 annual data report: epidemiology of kidney disease in the United States. *Am J Kidney Dis* (2015) 66(1 Suppl 1):S1–305. doi:10.1053/j.ajkd.2015.05.001
- Fick GM, Duley IT, Johnson AM, Strain JD, Manco-Johnson ML, Gabow PA. The spectrum of autosomal dominant polycystic kidney disease in children. *J Am Soc Nephrol* (1994) 4(9):1654–60.
- Fick-Brosnahan GM, Tran ZV, Johnson AM, Strain JD, Gabow PA. Progression of autosomal-dominant polycystic kidney disease in children. *Kidney Int* (2001) 59(5):1654–62. doi:10.1046/j.1523-1755.2001.0590051654.x
- Reed B, Nobakht E, Dadgar S, Bekheirnia MR, Masoumi A, Belibi F, et al. Renal ultrasonographic evaluation in children at risk of autosomal dominant polycystic kidney disease. *Am J Kidney Dis* (2010) 56(1):50–6. doi:10.1053/j. ajkd.2010.02.349
- 5. Zeier M, Geberth S, Schmidt KG, Mandelbaum A, Ritz E. Elevated blood pressure profile and left ventricular mass in children and young adults with

of mortality in ADPKD (71). Recent studies have shown that children and young adults with ADPKD also have vascular dysfunction as evidenced by impaired endothelial dependent dilation and increased arterial stiffness (72) as compared to healthy children. The results of such studies formed the rationale for a randomized placebo-controlled interventional trial of curcumin therapy in children and young adults with ADPKD (NCT02494141), which is currently ongoing. This study will assess brachial artery flowmediated dilation and aortic pulse wave velocity at baseline and following 12 months of curcumin vs. placebo. Secondary outcome measures include height-corrected TKV as well as various urine and plasma markers of inflammation and oxidative stress.

# CONCLUSION

Autosomal dominant polycystic kidney disease is a common kidney condition with major implications for long-term health. Structural kidney disease and vascular dysfunction are evident in childhood, and the earlier we can mitigate these processes, the more favorable the long-term renal and cardiovascular prognoses. It is critical to establish a reliable means to identify children at highest risk for rapid progression and to design safe interventional trials which can be initiated in childhood. Several previously investigated and potential therapeutic interventions have been reviewed here and their relevance to the pediatric ADPKD population discussed. Recent clinical trials suggest benefits of aggressive control of blood pressure with RAAS blockade as well as statin therapy in affected children. Tolvaptan shows great promise to mitigate structural and functional kidney disease progression in adults with advanced ADPKD, but cautious study is needed in affected children. The mechanisms of cyst formation are quite complex. Thus, it seems likely that a combination of treatments will be necessary to optimally inhibit cyst growth and that the ideal combination of therapies may vary over time relative to disease stage in both children and adults.

# **AUTHOR CONTRIBUTION**

The author confirms being the sole contributor of this work and approved it for publication.

autosomal dominant polycystic kidney disease. J Am Soc Nephrol (1993) 3(8):1451-7.

- Ivy DD, Shaffer EM, Johnson AM, Kimberling WJ, Dobin A, Gabow PA. Cardiovascular abnormalities in children with autosomal dominant polycystic kidney disease. J Am Soc Nephrol (1995) 5(12):2032–6.
- Cadnapaphornchai MA, Masoumi A, Strain JD, McFann K, Schrier RW. Magnetic resonance imaging of kidney and cyst volume in children with ADPKD. *Clin J Am Soc Nephrol* (2011) 6(2):369–76. doi:10.2215/ CJN.03780410
- O'Neill WC, Robbin ML, Bae KT, Grantham JJ, Chapman AB, Guay-Woodford LM, et al. Sonographic assessment of the severity and progression of autosomal dominant polycystic kidney disease: the Consortium of Renal Imaging Studies in Polycystic Kidney Disease (CRISP). Am J Kidney Dis (2005) 46(6):1058–64. doi:10.1053/j.ajkd.2005.08.026
- Klahr S, Levey AS, Beck GJ, Caggiula AW, Hunsicker L, Kusek JW, et al. The effects of dietary protein restriction and blood-pressure control on the progression of chronic renal disease. Modification of diet in Renal Disease Study Group. *N Engl J Med* (1994) 330(13):877–84. doi:10.1056/NEJM199403313301301

- Sarnak MJ, Greene T, Wang X, Beck G, Kusek JW, Collins AJ, et al. The effect of a lower target blood pressure on the progression of kidney disease: long-term follow-up of the modification of diet in renal disease study. *Ann Intern Med* (2005) 142(5):342–51. doi:10.7326/0003-4819-142-5-200503010-00009
- Group ET, Wuhl E, Trivelli A, Picca S, Litwin M, Peco-Antic A, et al. Strict blood-pressure control and progression of renal failure in children. *N Engl J Med* (2009) 361(17):1639–50. doi:10.1056/NEJMoa0902066
- Marlais M, Cuthell O, Langan D, Dudley J, Sinha MD, Winyard PJ. Hypertension in autosomal dominant polycystic kidney disease: a meta-analysis. Arch Dis Child (2016) 101(12):1142–7. doi:10.1136/archdischild-2015-310221
- Shamshirsaz AA, Reza Bekheirnia M, Kamgar M, Johnson AM, McFann K, Cadnapaphornchai M, et al. Autosomal-dominant polycystic kidney disease in infancy and childhood: progression and outcome. *Kidney Int* (2005) 68(5):2218–24. doi:10.1111/j.1523-1755.2005.00678.x
- Nowak KL, Cadnapaphornchai MA, Chonchol MB, Schrier RW, Gitomer B. Long-term outcomes in patients with very-early onset autosomal dominant polycystic kidney disease. *Am J Nephrol* (2016) 44(3):171–8. doi:10.1159/000448695
- Chapman AB, Johnson A, Gabow PA, Schrier RW. The renin-angiotensin-aldosterone system and autosomal dominant polycystic kidney disease. N Engl J Med (1990) 323(16):1091–6. doi:10.1056/NEJM199010183231602
- Schrier RW. Renal volume, renin-angiotensin-aldosterone system, hypertension, and left ventricular hypertrophy in patients with autosomal dominant polycystic kidney disease. J Am Soc Nephrol (2009) 20(9):1888–93. doi:10.1681/ASN.2008080882
- Cadnapaphornchai MA, McFann K, Strain JD, Masoumi A, Schrier RW. Prospective change in renal volume and function in children with ADPKD. *Clin J Am Soc Nephrol* (2009) 4(4):820–9. doi:10.2215/CJN.02810608
- Bender BG, Ellison MC, Gleason M, Murphy JR, Sundstrom DA, Szefler SJ. Minimizing attrition in a long-term clinical trial of pediatric asthma. Ann Allergy Asthma Immunol (2003) 91(2):168–76. doi:10.1016/ S1081-1206(10)62173-4
- Karlson CW, Rapoff MA. Attrition in randomized clinical trials for pediatric chronic conditions. *J Pediatr Psychol* (2009) 34(7):782–93. doi:10.1093/ jpepsy/jsn122
- Schrier RW, Abebe KZ, Perrone RD, Torres VE, Braun WE, Steinman TI, et al. Blood pressure in early autosomal dominant polycystic kidney disease. N Engl J Med (2014) 371(24):2255–66. doi:10.1056/NEJMoa1402685
- Johnson AM, Gabow PA. Identification of patients with autosomal dominant polcystic kidney disease at highest risk for end-stage renal disease. J Am Soc Nephrol (1997) 8(10):1560–7.
- 22. Chapman AB, Devuyst O, Eckardt K-U, Gansevoort RT, Harris T, Horie S, et al. Autosomal dominant polycystic kidney disease (ADPKD): executive summary from a kidney disease: improving global outcomes (KDIGO) controversies conference. *Kidney Int* (2015) 88(1):17–27. doi:10.1038/ki.2015.59
- National High Blood Pressure Education Program Working Group on High Blood Pressure in Children and Adolescents. The fourth report on the diagnosis, evaluation, and treatment of high blood pressure in children and adolescents. *Pediatrics* (2004) 114:555–76. doi:10.1542/peds.114.2.S2.555
- McFarlane SI, Muniyappa R, Francisco R, Sowers JR. Clinical review 145: pleiotropic effects of statins: lipid reduction and beyond. *J Clin Endocrinol Metab* (2002) 87(4):1451–8. doi:10.1210/jcem.87.4.8412
- Kostapanos MS, Liberopoulos EN, Elisaf MS. Statin pleiotropy against renal injury. J Cardiometab Syndr (2009) 4(1):E4–9. doi:10.1111/j.1559-4572.2008. 00052.x
- Zafar I, Tao Y, Falk S, McFann K, Schrier RW, Edelstein CL. Effect of statin and angiotensin-converting enzyme inhibition on structural and hemodynamic alterations in autosomal dominant polycystic kidney disease model. *Am J Physiol Renal Physiol* (2007) 293(3):F854–9. doi:10.1152/ajprenal.00059.2007
- van Dijk MA, Kamper AM, van Veen S, Souverijn JH, Blauw GJ. Effect of simvastatin on renal function in autosomal dominant polycystic kidney disease. *Nephrol Dial Transplant* (2001) 16(11):2152–7. doi:10.1093/ndt/16.11.2152
- Cadnapaphornchai MA, George DM, Masoumi A, McFann K, Strain JD, Schrier RW. Effect of statin therapy on disease progression in pediatric ADPKD: design and baseline characteristics of participants. *Contemp Clin Trials* (2011) 32(3):437–45. doi:10.1016/j.cct.2011.01.008
- Cadnapaphornchai MA, George DM, McFann K, Wang W, Gitomer B, Strain JD, et al. Effect of pravastatin on total kidney volume, left ventricular mass

index, and microalbuminuria in pediatric autosomal dominant polycystic kidney disease. *Clin J Am Soc Nephrol* (2014) 9(5):889–96. doi:10.2215/CJN.08350813

- Klawitter J, McFann K, Pennington AT, Wang W, Klawitter J, Christians U, et al. Pravastatin therapy and biomarker changes in children and young adults with autosomal dominant polycystic kidney disease. *Clin J Am Soc Nephrol* (2015) 10(9):1534–41. doi:10.2215/CJN.11331114
- Xu ZG, Yuan H, Lanting L, Li SL, Wang M, Shanmugam N, et al. Products of 12/15-lipoxygenase upregulate the angiotensin II receptor. J Am Soc Nephrol (2008) 19(3):559–69. doi:10.1681/ASN.2007080939
- Yu K, Bayona W, Kallen CB, Harding HP, Ravera CP, McMahon G, et al. Differential activation of peroxisome proliferator-activated receptors by eicosanoids. J Biol Chem (1995) 270(41):23975–83. doi:10.1074/jbc.270.41.23975
- Grantham JJ, Cook LT, Wetzel LH, Cadnapaphornchai MA, Bae KT. Evidence of extraordinary growth in the progressive enlargement of renal cysts. *Clin* J Am Soc Nephrol (2010) 5(5):889–96. doi:10.2215/CJN.00550110
- Grantham JJ, Mulamalla S, Swenson-Fields KI. Why kidneys fail in autosomal dominant polycystic kidney disease. *Nat Rev Nephrol* (2011) 7(10):556–66. doi:10.1038/nrneph.2011.109
- 35. Grantham JJ. Rationale for early treatment of polycystic kidney disease. *Pediatr Nephrol* (2015) 30(7):1053–62. doi:10.1007/s00467-014-2882-8
- 36. Grantham JJ, Mangoo-Karim R, Uchic ME, Grant M, Shumate WA, Park CH, et al. Net fluid secretion by mammalian renal epithelial cells: stimulation by cAMP in polarized cultures derived from established renal cells and from normal and polycystic kidneys. *Trans Assoc Am Physicians* (1989) 102:158–62.
- Mangoo-Karim R, Uchic ME, Grant M, Shumate WA, Calvet JP, Park CH, et al. Renal epithelial fluid secretion and cyst fluid growth: the role of cyclic AMP. FASEB J (1989) 3(14):2629–32.
- Torres VE, Wang X, Qian Q, Somlo S, Harris PC, Gattone VH II. Effective treatment of an orthologous model of autosomal dominant polycystic kidney disease. *Nat Med* (2004) 10(4):363–4. doi:10.1038/nm1004
- Wang X, Gattone V II, Harris PC, Torres VE. Effectiveness of vasopressin V2 receptor antagonists OPC-31260 and OPC-41061 on polycystic kidney disease development in the PCK rat. J Am Soc Nephrol (2005) 16(4):846–51. doi:10.1681/ASN.2004121090
- Torres VE, Meijer E, Bae KT, Chapman AB, Devuyst O, Gansevoort RT, et al. Rationale and design of the TEMPO (tolvaptan efficacy and safety in management of autosomal dominant polycystic kidney disease and its outcomes) 3-4 study. Am J Kidney Dis (2011) 57(5):692–9. doi:10.1053/j.ajkd.2010.11.029
- Torres VE, Chapman AB, Devuyst O, Gansevoort RT, Grantham JJ, Higashihara E, et al. Tolvaptan in patients with autosomal dominant polycystic kidney disease. N Engl J Med (2012) 367(25):2407–18. doi:10.1056/NEJMoa1205511
- Torres VE, Higashihara E, Devuyst O, Chapman AB, Gansevoort RT, Grantham JJ, et al. Effect of tolvaptan in autosomal dominant polycystic kidney disease by CKD stage: results from the TEMPO 3:4 trial. *Clin J Am Soc Nephrol* (2016) 11(5):803–11. doi:10.2215/CJN.06300615
- 43. Gansevoort RT, Arici M, Benzing T, Birn H, Capasso G, Covic A, et al. Recommendations for the use of tolvaptan in autosomal dominant polycystic kidney disease: a position statement on behalf of the ERA-EDTA working groups on inherited kidney disorders and the European renal best practice. *Nephrol Dial Transplant* (2016) 31(3):337–48. doi:10.1093/ndt/gfv456
- Erickson KF, Chertow GM, Goldhaber-Fiebert JD. Cost-effectiveness of tolvaptan in autosomal dominant polycystic kidney disease. *Ann Intern Med* (2013) 159:382–9. doi:10.7326/0003-4819-159-6-201309170-00004
- Wang CJ, Creed C, Winklhofer FT, Grantham JJ. Water prescription in autosomal dominant polycystic kidney disease: a pilot study. *Clin J Am Soc Nephrol* (2011) 6(1):192–7. doi:10.2215/CJN.03950510
- 46. Amro OW, Paulus JK, Noubary F, Perrone RD. Low-osmolar diet and adjusted water intake for vasopressin reduction in autosomal dominant polycystic kidney disease: a pilot randomized controlled trial. *Am J Kidney Dis* (2016) 68(6):882–91. doi:10.1053/j.ajkd.2016.07.023
- Higashihara E, Nutahara K, Tanbo M, Hara H, Miyazaki I, Kobayashi K, et al. Does increased water intake prevent disease progression in autosomal dominant polycystic kidney disease? *Nephrol Dial Transplant* (2014) 29(9):1710–9. doi:10.1093/ndt/gfu093
- 48. Caroli A, Perico N, Perna A, Antiga L, Brambilla P, Pisani A, et al. Effect of longacting somatostatin analogue on kidney and cyst growth in autosomal dominant polycystic kidney disease (ALADIN): a randomised,

placebo-controlled, multicentre trial. Lancet (2013) 382(9903):1485-95. doi:10.1016/S0140-6736(13)61407-5

- Pisani A, Sabbatini M, Imbriaco M, Riccio E, Rubis N, Prinster A, et al. Long-term effects of octreotide on liver volume in patients with polycystic kidney and liver disease. *Clin Gastroenterol Hepatol* (2016) 14(7):1022–30. doi:10.1016/j.cgh.2015.12.049
- Gevers TJ, Hol JC, Monshouwer R, Dekker HM, Wetzels JF, Drenth JP. Effect of lanreotide on polycystic liver and kidneys in autosomal dominant polycystic kidney disease: an observational trial. *Liver Int* (2015) 35(5):1607–14. doi:10.1111/liv.12726
- 51. Shillingford JM, Murcia NS, Larson CH, Low SH, Hedgepeth R, Brown N, et al. The mTOR pathway is regulated by polycystin-1, and its inhibition reverses renal cystogenesis in polycystic kidney disease. *Proc Natl Acad Sci U* S A (2006) 103(14):5466–71. doi:10.1073/pnas.0509694103
- Zafar I, Ravichandran K, Belibi FA, Doctor RB, Edelstein CL. Sirolimus attenuates disease progression in an orthologous mouse model of human autosomal dominant polycystic kidney disease. *Kidney Int* (2010) 78(8):754–61. doi:10.1038/ki.2010.250
- Renken C, Fischer DC, Kundt G, Gretz N, Haffner D. Inhibition of mTOR with sirolimus does not attenuate progression of liver and kidney disease in PCK rats. *Nephrol Dial Transplant* (2011) 26(1):92–100. doi:10.1093/ndt/gfq384
- Serra AL, Poster D, Kistler AD, Krauer F, Raina S, Young J, et al. Sirolimus and kidney growth in autosomal dominant polycystic kidney disease. N Engl J Med (2010) 363(9):820–9. doi:10.1056/NEJMoa0907419
- Walz G, Budde K, Mannaa M, Nurnberger J, Wanner C, Sommerer C, et al. Everolimus in patients with autosomal dominant polycystic kidney disease. N Engl J Med (2010) 363(9):830–40. doi:10.1056/NEJMoa1003491
- Zou MH, Kirkpatrick SS, Davis BJ, Nelson JS, Wiles WG IV, Schlattner U, et al. Activation of the AMP-activated protein kinase by the anti-diabetic drug metformin in vivo. Role of mitochondrial reactive nitrogen species. *J Biol Chem* (2004) 279(42):43940–51. doi:10.1074/jbc.M404421200
- Takiar V, Nishio S, Seo-Mayer P, King JD Jr, Li H, Zhang L, et al. Activating AMP-activated protein kinase (AMPK) slows renal cystogenesis. *Proc Natl* Acad Sci U S A (2011) 108(6):2462–7. doi:10.1073/pnas.1011498108
- Ewart MA, Kennedy S. AMPK and vasculoprotection. *Pharmacol Ther* (2011) 131(2):242–53. doi:10.1016/j.pharmthera.2010.11.002
- Sharma S, Kulkarni SK, Chopra K. Curcumin, the active principle of turmeric (*Curcuma longa*), ameliorates diabetic nephropathy in rats. *Clin Exp Pharmacol Physiol* (2006) 33(10):940–5. doi:10.1111/j.1440-1681.2006.04468.x
- Bayrak O, Uz E, Bayrak R, Turgut F, Atmaca AF, Sahin S, et al. Curcumin protects against ischemia/reperfusion injury in rat kidneys. World J Urol (2008) 26(3):285–91. doi:10.1007/s00345-008-0253-4
- Ghosh SS, Massey HD, Krieg R, Fazelbhoy ZA, Ghosh S, Sica DA, et al. Curcumin ameliorates renal failure in 5/6 nephrectomized rats: role of inflammation. *Am J Physiol Renal Physiol* (2009) 296(5):F1146–57. doi:10.1152/ ajprenal.90732.2008
- 62. LI R, Wang Y, Liu Y, Chen Q, Fu W, Wang H, et al. Curcumin inhibits transforming growth factor-beta1-induced EMT via PPARgamma pathway, not Smad pathway in renal tubular epithelial cells. *PLoS One* (2013) 8(3):e58848. doi:10.1371/journal.pone.0058848

- Zhou X, Zhang J, Xu C, Wang W. Curcumin ameliorates renal fibrosis by inhibiting local fibroblast proliferation and extracellular matrix deposition. *J Pharmacol Sci* (2014) 126(4):344–50. doi:10.1254/jphs.14173FP
- Gao J, Zhou H, Lei T, Zhou L, Li W, Li X, et al. Curcumin inhibits renal cyst formation and enlargement in vitro by regulating intracellular signaling pathways. *Eur J Pharmacol* (2011) 654(1):92–9. doi:10.1016/j.ejphar.2010.12.008
- Leonhard WN, van der Wal A, Novalic Z, Kunnen SJ, Gansevoort RT, Breuning MH, et al. Curcumin inhibits cystogenesis by simultaneous interference of multiple signaling pathways: in vivo evidence from a Pkd1-deletion model. *Am J Physiol Renal Physiol* (2011) 300(5):F1193–202. doi:10.1152/ ajprenal.00419.2010
- 66. Nakmareong S, Kukongviriyapan U, Pakdeechote P, Kukongviriyapan V, Kongyingyoes B, Donpunha W, et al. Tetrahydrocurcumin alleviates hypertension, aortic stiffening and oxidative stress in rats with nitric oxide deficiency. *Hypertens Res* (2012) 35(4):418–25. doi:10.1038/hr.2011.180
- Rungseesantivanon S, Thenchaisri N, Ruangvejvorachai P, Patumraj S. Curcumin supplementation could improve diabetes-induced endothelial dysfunction associated with decreased vascular superoxide production and PKC inhibition. BMC Complement Altern Med (2010) 10:57. doi:10.1186/1472-6882-10-57
- Majithiya JB, Balaraman R. Time-dependent changes in antioxidant enzymes and vascular reactivity of aorta in streptozotocin-induced diabetic rats treated with curcumin. J Cardiovasc Pharmacol (2005) 46(5):697–705. doi:10.1097/01. fjc.0000183720.85014.24
- Fleenor BS, Sindler AL, Marvi NK, Howell KL, Zigler ML, Yoshizawa M, et al. Curcumin ameliorates arterial dysfunction and oxidative stress with aging. *Exp Gerontol* (2013) 48(2):269–76. doi:10.1016/j.exger.2012.10.008
- Wang D, Iversen J, Wilcox CS, Strandgaard S. Endothelial dysfunction and reduced nitric oxide in resistance arteries in autosomal-dominant polycystic kidney disease. *Kidney Int* (2003) 64(4):1381–8. doi:10.1046/j.1523-1755.2003.00236.x
- Fick GM, Johnson AM, Hammond WS, Gabow PA. Causes of death in autosomal dominant polycystic kidney disease. J Am Soc Nephrol (1995) 5(12):2048–56.
- Nowak KL, Farmer H, Cadnapaphornchai MA, Gitomer B, Chonchol M. Vascular dysfunction in children and young adults with autosomal dominant polycystic kidney disease. *Nephrol Dial Transplant* (2017) 32(2):342–7. doi:10.1093/ndt/gfw013

**Conflict of Interest Statement:** The author declares that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Copyright © 2017 Cadnapaphornchai. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) or licensor are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.